Author's response to reviews

Title: Physical rehabilitation interventions for adult patients with critical illness across the continuum of recovery: an overview of systematic reviews protocol

Authors:

- Bronwen Connolly (bronwen.connolly@nhs.net)
- Brenda O’Neill (b.oneill@ulster.ac.uk)
- Lisa Salisbury (lsalisbu@exseed.ed.ac.uk)
- Kathryn McDowell (mcdowell-k3@email.ulster.ac.uk)
- Bronagh Blackwood (b.blackwood@qub.ac.uk)

Version: 3
Date: 16 September 2015

Author's response to reviews: see over
Dear Dr Moher,

Re: Re-submission of revised systematic review protocol

MS: 3512724761760713

‘Physical rehabilitation interventions for adult patients with critical illness across the continuum of recovery: an overview of systematic reviews protocol’

On behalf of all the authors, our thanks for the decision by the Editor-in-Chief to accept our manuscript for publication pending abstract formatting corrections. I can confirm that this has been completed and provide a point-by-point response below, with changes highlighted in the text for enhanced clarification.

We hope the manuscript is now formatted accordingly for the Journal and extend our thanks once again.

Yours sincerely

Dr Bronwen Connolly BSc (Hons), MSc, PhD, MCSP
Senior Clinical Research Physiotherapist
Lane Fox Respiratory Unit
St Thomas’ Hospital
Guy’s & St Thomas’ NHS Foundation Trust
Westminster Bridge Road
London
SE1 7EH
United Kingdom

E-mail: bronwen.connolly@nhs.net
Tel: 020 7188 8070
Response to the Editor’s comments

Editorial request 1:

1) Please ensure that your abstract is correctly formatted. This should not exceed 350 words and should be structured into separate sections headed Background, Methods/Design, Discussion (if appropriate). Please minimize the use of abbreviations and do not cite references in the abstract. Systematic review registration, if your reports the results of a controlled health care intervention, please list your registry, along with the unique identifying number (e.g. Systematic review registration: PROSPERO CRD0123456789). Please note that there should be no space between the letters and numbers of your registration number.

Response

Many thanks for this comment and our sincere apologies for the oversight in missing some adjustments to the re-formatting of the abstract in our initial revision. We have clarified section headings and also the reporting of the systematic review registration.